Optimal Dynamic Trading Strategies with Risk Limits

E

Domenico CUOCO The Wharton School, University of Pennsylvania

Hua HE School of Management, Yale University

Sergei ISSAENKO The Wharton School, University of Pennsylvania

Winner of the 2002 FAME Research Prize Research Paper N° 60

FAME - International Center for Financial Asset Management and Engineering



Research Paper Series

The International Center for Financial Asset Management and Engineering (FAME) is a private foundation created in 1996 at the initiative of 21 leading partners of the finance and technology community together with three Universities of the Lake Geneva Region (Universities of Geneva, University of Lausanne and the Graduate Institute of International Studies).

Fame is about *research*, *doctoral training*, and *executive education* with "interfacing" activities such as the FAME lectures, the Research Day/Annual Meeting, and the Research Paper Series.

The FAME Research Paper Series includes three types of contributions:

- First, it reports on the research carried out at FAME by students and research fellows.
- Second, it includes research work contributed by Swiss academics and practitioners interested in a wider dissemination of their ideas, in practitioners' circles in particular.
- Finally, prominent international contributions of particular interest to our constituency are included as well on a regular basis.

FAME will strive to promote the research work in finance carried out in the three partner Universities. These papers are distributed with a 'double' identification: the FAME logo and the logo of the corresponding partner institution. With this policy, we want to underline the vital lifeline existing between FAME and the Universities, while simultaneously fostering a wider recognition of the strength of the academic community supporting FAME and enriching the Lemanic region.

Each contribution is preceded by an Executive Summary of two to three pages explaining in nontechnical terms the question asked, discussing its relevance and outlining the answer provided. We hope the series will be followed attentively by all academics and practitioners interested in the fields covered by our name.

I am delighted to serve as coordinator of the FAME Research Paper Series. Please contact me if you are interested in submitting a paper or for all suggestions concerning this matter.

Sincerely,

Martin E. R. M.

Prof. Martin Hoesli University of Geneva, HEC 40 bd du Pont d'Arve 1211 Genève 4 Tel: +41 (022) 705 8122 Martin.Hoesli@hec.unige.ch

Winner of the 2002 FAME Research Prize

OPTIMAL DYNAMIC TRADING STRATEGIES WITH RISK LIMITS

Domenico CUOCO Hua HE Sergei ISSAENKO

Optimal Dynamic Trading Strategies with Risk Limits^{*}

Domenico Cuoco The Wharton School University of Pennsylvania Philadelphia, PA 19104 cuoco@wharton.upenn.edu Hua He School of Management Yale University New Haven, CT 06520 hua.he@yale.edu

Sergei Issaenko The Wharton School University of Pennsylvania Philadelphia, PA 19104 serguei@wharton.upenn.edu

First draft: April 2001 This draft: December 2001

 $^{^{\}ast}\mathrm{We}$ are grateful to Carlo Acerbi and seminar participants at CalTech, Wharton and Yale for helpful comments.

Abstract

Value at Risk (VaR) has emerged in recent years as a standard tool to measure and control the risk of trading portfolios. Yet, existing theoretical analyses of the optimal behavior of a trader subject to VaR limits have produced a negative view of VaR as a risk-control tool. In particular, VaR limits have been found to induce increased risk exposure in some states and an increased probability of extreme losses. However, these conclusions are based on models that are either static or dynamically inconsistent. In this paper we formulate a dynamically consistent model of optimal portfolio choice subject to VaR limits and show that the conclusions of earlier papers are incorrect if, consistently with common practice, the portfolio VaR is reevaluated dynamically making use of available conditioning information. In particular, we find that the risk exposure of a trader subject to a VaR limit is always lower than that of an unconstrained trader and that the probability of extreme losses is also lower. We also consider risk limits formulated in terms of Tail Conditional Expectation (TCE), a coherent risk measure often advocated as an alternative to VaR, and show that in our dynamic setting it is always possible to transform a TCE limit into an equivalent VaR limit, and conversely. Journal of Economic Literature Classification Numbers: D91, D92, G11, C61.

Executive Summary

Optimal Dynamic Trading Strategies with Risk Limits

Domenico Cuoco Hua He Sergei Issaenko

Investment firms customarily limit the discretionality of their traders by imposing limits on the risk of trading portfolios. Since Value at Risk (VaR) has gained in recent years increasing popularity as a risk measure, these limits are frequently specified in terms of VaR. Some recent academic papers have however argued that VaR suffers serious shortcomings as a risk-control tool. In particular, since VaR measures are insensitive to the expected value of losses in the tail of the distribution, traders subject to a VaR limit might have the incentive to post very large losses in the exceptional cases where losses exceed the VaR limit and to increase the risk of their portfolios in some states. For this reason, Tail Conditional Expectation (TCE) is often advocated as a better risk-control tool.

However, these conclusions are based on models that are either entirely static, or, even when they allow for dynamic trading, invariably assume that the risk of the portfolio is not reevaluated dynamically as the composition of the portfolio changes.

Our first contribution in this paper is to develop a more realistic dynamic model of the optimal behavior of a trader subject to risk constraints. In particular, we assume that the risk of the trading portfolio is reevaluated dynamically: thus, the trader must satisfy the specified risk limit *at all times*, rather than only at the initial date. In addition, we make the risk computations in our model consistent with practice by assuming that, when assessing the risk of a portfolio, the distribution of the portfolio profits and losses at the chosen horizon is computed assuming that the current portfolio composition is kept unchanged over this horizon. In our dynamic setting, we also allow the risk limit to vary as a function of the portfolio value and time and examine the behavior of optimal trading strategies under alternative functional specifications of this limit.

When portfolio risk is measured by VaR, the optimal trading behavior under the dynamic risk limit described above is significantly different from that implied by a static VaR constraint and results in a much more favorable assessment of VaR as a risk-control tool. In particular, we find that the proportional allocation to risky assets is always lower than what it would have been in the absence of the VaR constraint and that the probability of extreme losses is never larger than what it would have been in the absence of the constraint. Optimal investment strategies still display two-fund separation (with the two funds being the riskless asset and the instantaneous mean-variance efficient portfolio): this result remains true even if the VaR limit is allowed to vary as the value of the trading portfolio changes. Thus, a dynamic VaR constraint does not distort the composition of the optimal portfolio of risky assets: instead, it simply impacts the relative allocation to the riskless and the risky fund.

We also consider the optimal behavior of a trader subject to a TCE limit and prove that in our setting TCE and VaR are equivalent as risk-control tools: specifically, given a dynamic TCE limit, it is always possible to identify a dynamic VaR limit that would induce the same optimal trading strategy (irrespective of the trader's preferences), and conversely. This is true in spite of the fact that TCE is a coherent risk measure, while VaR is not, and results from the fact that VaR, while not being subadditive, is comonotone additive (in the sense of Pflug (2000)): in our setting, the comonotonicity property arises naturally for optimal portfolios as a result of two-fund separation.

These findings provide some theoretical support for the growing use of VaR as a risk-control tool.

1. Introduction

Investment firms customarily limit the discretionality of their traders by imposing limits on the risk of trading portfolios. Since Value at Risk (VaR) has gained in recent years increasing popularity as a risk measure,¹ these limits are frequently specified in terms of VaR.²

The popularity of VaR is due at least in part to the fact that it is an easily-understood measure of risk: specifically, VaR is the maximum loss of a portfolio over a given horizon, at a given confidence level. The choice of a horizon and confidence level are largely arbitrary, although the Basle Committee proposals of April 1995 prescribed that VaR computations for the purpose of assessing bank capital requirements should be based on a uniform horizon of 10 trading days (two calendar weeks) and a 99% confidence level.³ The use of VaR as a risk measure has been endorsed by regulators and industry groups, including the Basle Committee on Banking Supervision, the SEC, the Group of Thirty (an international consultative group of leading bankers, financiers and academics), the International Swap and Derivatives Association (which represents more than 150 leading financial institutions dealing in privately negotiated over-the-counter derivatives transactions) and the Derivatives Policy Group (which comprises the six U.S. brokers-dealers with the largest OTC derivatives affiliates).⁴ Both J.P. Morgan and Bankers Trust have introduced risk management systems (called RiskMetrics and RAROC 2020, respectively) that produce VaR measures.

In spite of its widespread acceptance, VaR is also known to possess unappealing features. Artzner, Delbaen, Eber and Heath (1999) proposed an axiomatic foundation of risk measures, by identifying four properties that a reasonable risk measure should satisfy and providing a characterization of the risk measures satisfying these properties, which they called *coherent risk measures*. VaR is not a coherent risk measure, as it does not satisfy the subadditivity property: in other words, the VaR associated with a combination of two portfolios can be higher than the sum of the VaRs of the two individual portfolios. This has induced Artzner, Delbaen, Eber and Heath to propose the use of the Tail Conditional expectation (TCE), defined as the conditional expectation of losses above the VaR, as an

¹See "The Risk Business", The Economist, October 17, 1998.

²As noted by Jorion (2001, p. 379), "At the business area or unit level, VaR [...] can be used to set position limits for traders and to decide where to allocate limited capital resources. A great advantage of VaR is that it creates a common denominator with which to compare various risky activities. Traditionally, position limits are set in terms of notional exposure. A trader, for instance, may have a limit of \$10 million on overnight positions in 5-year Treasuries. The same limit for 30-year Treasuries or in Treasury bond futures, however, is substantially riskier. Thus, notional position limits are not directly comparable across units. Instead, VaR provides a common denominator to compare various asset classes and can be used as a guide to set position limits for business units." As it will become clear in the following analysis, the risk limits we consider translate naturally into position limits that take into account both the risk and the expected return of the position (see Remarks 2 and 3).

³Banks using a 1-day horizon for internal VaR reporting are allowed to obtain their 10-day VAR by simply multiplying the 1-day VaR by the square root of 10: see Jorion (2001, pp. 64–65).

⁴See Jorion (2001, pp. 43–49).

alternative to VaR.⁵ Artzner, Delbaen, Eber and Heath proved that TCE is a coherent risk measure under a technical condition on the risk distribution. Pflug (2000) provides a general proof of coherence for continuous risk distributions and discusses several additional desirable properties of TCE (see also Embrechts (1999)).

Our focus in this paper is on the dynamic portfolio choice of a trader subject to a risk limit specified in terms of VaR or TCE. This problem has not yet received a complete treatment in the existing literature.⁶ Ahn, Boudoukh, Richardson and Whitelaw (1999) study the static minimization of the VaR of a given stock exposure using put options on the stock. Alexander and Baptista (2000), Huisman, Koedijk and Pownall (1999), Kast, Luciano and Peccati (1999) and Vorst (2001), among others, focus on the maximization of the expected return of a portfolio subject to a VaR constraint in a static (one-period) setting, while Rockafellar and Uryasev (2001) consider TCE minimization, again in a static setting. To our knowledge, the only analyses of portfolio choice subject to risk limits in models with dynamic trading are in two recent papers by Emmer, Klüppelberg and Korn (2001) and Basak and Shapiro (2001).⁷

Emmer, Klüppelberg and Korn consider a model with continuous trading in which traders face a VaR limit. However, for analytical tractability, they only consider strategies that maintain fixed portfolio weights: this reduces their problem to a static one and results in a dynamically-inconsistent trading strategy.

Basak and Shapiro consider the following static optimization problem:

s.t.
$$\begin{array}{c} \max_{W_T \ge 0} \mathbf{E}[u(W_T)] \\ \mathbf{E}[\xi_T W_T] \le W_0, \\ P[W_0 - W_T > \overline{VaR}] \le \alpha, \end{array}$$
(1)

where u is the trader's utility function, T > 0 is the investment horizon (which is assumed to coincide with the VaR horizon), W_T (respectively, $W_0 > 0$) is the terminal (respectively, the initial) portfolio value, ξ_T is the state-price density at time $T, 1-\alpha \in (0,1)$ is the chosen confidence level and $\overline{VaR} \ge 0$. The first constraint in (1) is the usual budget constraint, while the second constraint is equivalent to the portfolio VaR being no larger than \overline{VaR} . The problem in (1) is interpreted as the static formulation of a dynamic portfolio problem subject to a VaR constraint at time 0 in a complete-market economy with continuous trading. Letting $\overline{\xi}$ be such that $P[\xi_T > \overline{\xi}] = \alpha$, Basak and Shapiro show that, whenever the constraint is binding, a trader forced to reduce portfolio losses in some states to satisfy the VaR constraint would optimally choose to finance these reduced losses by increasing the portfolio losses in the "costly states" where $\xi_T > \overline{\xi}$. Since these states are already the ones with the lowest terminal portfolio value under the unconstrained optimal policy, the VaR constraint results in a fattening of the left tail of the distribution of the terminal portfolio value (i.e., in an increased probability of extreme losses). This leads Basak and Shapiro to conclude:

⁵TCE is also sometimes referred to as Conditional VaR, Tail VaR, Mean Excess Loss, Conditional Loss or Tail Loss.

⁶For a review of the literature on VaR and related risk measures, see the book of Jorion (2001), the review article of Duffie and Pan (1997) and the extensive on-line references at www.gloriamundi.org.

⁷Basak and Shapiro (2001) also contains an analysis of the general equilibrium implications of VaR limits.

The [VaR risk-management] is viewed by many as a tool to shield economic agents from large losses, which, when they occur, could cause credit and solvency problems. But our solution reveals that when a large loss occurs, it is a yet larger loss under the [VaR risk-management] and hence more likely to lead to credit problems, defeating the very purpose of using the [VaR risk-management]. (p. 378)⁸

Not surprisingly, Basak and Shapiro also find that, with lognormally-distributed returns, the VaR constraint in problem (1) induces traders to invest significantly more in risky assets in some states than they would have invested in the absence of the constraint: this increase in risk-taking is necessary to realize increased losses in the "costly states". Finally, Basak and Shapiro report that a risk limit specified in terms of a tail-expectation-based measure would result in neither an increased probability of extreme losses nor in an increased allocation to risky assets in some states: thus, tail-expectation-based measures should be preferred to quantile-based measures (such as VaR) for the purpose of risk control.⁹

However, the problem in (1) has several shortcomings as a model of the dynamic portfolio choice of a trader subject to risk limits. First, it assumes that the portfolio's VaR is never reevaluated after the initial date: thus, the conditional probability of portfolio losses below the prescribed maximum VaR can become zero after the initial date and yet the trader is allowed to continue to follow his trading strategy. This assumption is extreme: in practice, most financial institutions using VaR for internal risk control reevaluate it at least daily.¹⁰ Second, because the VaR limit is only imposed at the initial date, the trading strategy solving (1) is dynamically inconsistent and must be interpreted as a commitment solution: otherwise, the trader would find it optimal to instantaneously revert to the unconstrained-optimal investment strategy after the initial date and the VaR constraint would never be binding. Third, the formulation in (1) assumes that the portfolio VaR is computed under full knowledge of the trader's behavior in all future contingencies. Again, this assumption is extreme and it does not match actual practice: as noted by Jorion (2001, p. 107), VaR is invariably computed under the assumption that the existing portfolio is kept unchanged over the VaR horizon.¹¹

⁸Similarly, Vorst (2001) states: "Recently, financial institutions discovered that portfolios with a limited Value at Risk often showed returns that were close to the VaR and had large losses in the exceptional cases where losses exceeded VaR. [The] theoretically optimal portfolios indeed have the properties as experienced by financial institutions and illustrate that maximizing under a VaR-constraint is very dangerous."

⁹Basak and Shapiro consider a tail-expectation-based risk measure, which they call *Limited-Expected-Losses* (LEL), that is computed under the equivalent martingale measure rather than under the actual probability measure.

¹⁰The Basle Committee proposals of April 1995 require banks to recompute the VaR of their portfolios on a daily basis, as capital requirements are proportional to the higher of the previous day's VaR or the average VaR over the last 60 business days: see Jorion (2001, p. 64). Similarly, the 1993 "Best Practices" Recommendations from the Group of Thirty stated that "Dealers should use a consistent measure to calculate *daily* the market risk of their position, which is best measured with a Value-at-Risk (VaR) approach." (Jorion, 2001, p. 485).

¹¹Because VaR measures are computed under the assumption that the existing portfolio is kept unchanged over the VaR horizon, companies normally select a VaR horizon for internal reporting purposes taking into account the turnover of the trading portfolio. Accordingly, "Commercial banks currently report their trading VaR over a daily horizon because of the liquidity and rapid turnover in their portfolios. In contrast, investment portfolios such as pension funds generally invest in less liquid assets and adjust their risk exposures

Our first contribution in this paper is to develop a more realistic dynamically-consistent model of the optimal behavior of a trader subject to risk constraints. Differently from Basak and Shapiro, we assume that the risk of the trading portfolio is reevaluated dynamically (in fact, continuously), making full use of conditioning information: thus, the trader must satisfy the specified risk limit *at all times*, rather than only at the initial date. In addition, we make the risk computations in our model consistent with practice by assuming that, when assessing the risk of a portfolio, the distribution of the portfolio value at the chosen horizon is computed assuming that the current portfolio composition is kept unchanged over this horizon. For technical convenience, we restrict ourselves to the case of lognormallydistributed returns.

When risk is measured by VaR, the optimal trading behavior under the dynamic risk limit described above is significantly different from that implied by the static VaR constraint of Basak and Shapiro and results in a much more favorable assessment of VaR as a riskcontrol tool. In particular, we find that the proportional allocation to risky assets is always lower than what it would have been in the absence of VaR the constraint and that the probability of extreme losses is always no larger than what it would have been in the absence of the constraint. As in Basak and Shapiro, we find that the optimal investment strategy still displays two-fund separation, the two funds being the riskless asset and the instantaneous mean-variance efficient portfolio. Thus, a dynamically-reevaluated VaR constraint does not distort the composition of the optimal portfolio of risky assets: instead, it simply impacts the relative allocation to the riskless and the risky fund.

We also consider the optimal behavior of a trader subject to a TCE limit and prove that in our setting TCE and VaR are equivalent as risk-control tools: specifically, given a dynamic TCE limit, it is always possible to identify a dynamic VaR limit that would induce the same optimal trading strategy (irrespective of the trader's preferences), and conversely. This is true in spite of the fact that TCE is a coherent risk measure, while VaR is not, and results from the fact that VaR, while not being subadditive, is comonotone additive (in the sense of Pflug (2000)):¹² in our setting, the comonotonicity property arises naturally for optimal portfolios as a result of two-fund separation.

The rest of the paper is organized as follows. Section 2 describes our model. Section 3 contains the main characterization result of optimal trading strategies under VaR constraints. Section 4 provides some explicit examples with CRRA utilities. Section 5 considers the case of TCE constraints and establishes the equivalence result. Section 6 concludes and an Appendix contains all the proofs.

2. The Model

We consider a continuous-time stochastic economy on the finite horizon [0, T]. Uncertainty is represented by a filtered probability space $(\Omega, \mathcal{F}, \mathbf{F}, P)$, where $\mathbf{F} = \{\mathcal{F}_t\}$ is the natural filtration generated by a *d*-dimensional Brownian motion w.

only slowly, which is why a 1-month horizon is generally chosen for investment purposes." (Jorion, 2001, p. 117).

¹²Two random variables X and Y on the same probability space (Ω, \mathcal{F}, P) are said to be *comonotone* if $(X(\omega_1) - X(\omega_2))(Y(\omega_1) - Y(\omega_2)) \ge 0$ a.s. for all $\omega_1, \omega_2 \in \Omega$.

The investment opportunities are represented by n+1 long-lived securities. The first security (the "bond") is a money market account earning a constant continuously-compounded interest rate r > 0. The other n assets (the "stocks") are risky and their price process S(inclusive of reinvested dividends) is a n-dimensional geometric Brownian motion with drift vector $r\bar{1} + \mu$ and diffusion matrix σ , i.e.,

$$S_t = S_0 + \int_0^t I_s^S(r\bar{1} + \mu) \, ds + \int_0^t I_s^S \sigma \, dw_s,$$

where I_t^S denotes the $n \times n$ diagonal matrix with elements S_t and $\overline{1} = (1, \ldots, 1)^{\top}$. We assume without loss of generality that $1 \leq n \leq d$ and that $\operatorname{rank}(\sigma) = n$.¹³ Trading in the bond and in the stocks takes place continuously and is frictionless. An admissible trading strategy is an adapted *n*-dimensional portfolio-weight process π with $\int_0^T |\pi_t|^2 ds < \infty$.¹⁴ Let Π denote the set of admissible trading strategies. Given a trading strategy $\pi \in \Pi$, the associated portfolio value process W^{π} satisfies the dynamic budget constraint

$$W_{t}^{\pi} = W_{0} + \int_{0}^{t} W_{s}^{\pi} \left(r + \pi_{s}^{\top} \mu \right) \, ds + \int_{0}^{t} W_{s}^{\pi} \pi_{s}^{\top} \sigma \, dw_{s},$$

$$T = W_{0} \exp\left(\int_{0}^{t} \left(r + \pi_{s}^{\top} \mu - \frac{1}{2} |\pi_{s}^{\top} \sigma|^{2} \right) \, ds + \int_{0}^{t} \pi_{s}^{\top} \sigma \, dw_{s} \right), \tag{2}$$

or

$$W_t^{\pi} = W_0 \exp\left(\int_0^t \left(r + \pi_s^{\mathsf{T}} \mu - \frac{1}{2} |\pi_s^{\mathsf{T}} \sigma|^2\right) \, ds + \int_0^t \pi_s^{\mathsf{T}} \sigma \, dw_s\right),\tag{2}$$

where $W_0 > 0$ denotes the initial value of the portfolio. Notice that (2) implies

$$W_{t+\tau}^{\pi} = W_t^{\pi} \exp\left(\int_t^{t+\tau} \left(r + \pi_s^{\top} \mu - \frac{1}{2} |\pi_s^{\top} \sigma|^2\right) ds + \int_t^{t+\tau} \pi_s^{\top} \sigma dw_s\right)$$
(3)

for any $\tau > 0$.

For given $\tau > 0$, W > 0 and $\pi \in \mathbb{R}^n$, let

$$\mathcal{W}_{t+\tau}(W,\pi) = W \exp\left(\left(r + \pi^{\top} \mu - \frac{1}{2} |\pi^{\top} \sigma|^2\right) \tau + \pi^{\top} \sigma(w_{t+\tau} - w_t)\right).$$

It follows immediately from (3) that, given a portfolio π_t and the associated portfolio value W_t^{π} at time t, the random variable $\mathcal{W}_{t+\tau}(W_t^{\pi}, \pi_t)$ would be the future value of the portfolio at time $t + \tau$ if the portfolio weights were kept constant between time t and time $t + \tau$.

For a given probability level $\alpha \in (0, 1)$ and a given horizon $\tau > 0$, the VaR at time t of a portfolio $\pi \in \Pi$, denoted by $VaR_t^{\alpha, \pi}$, is then given by

$$VaR_{t}^{\alpha,\pi} = \inf \left\{ L \ge 0 : P\left(W_{t}^{\pi} - \mathcal{W}_{t+\tau}(W_{t}^{\pi}, \pi_{t}) \ge L \mid \mathcal{F}_{t}\right) < \alpha \right\} = (Q_{t}^{\alpha,\pi})^{-},$$
(4)

where

$$Q_t^{\alpha,\pi} = \sup \left\{ L \in \mathbb{R} : P\left(\mathcal{W}_{t+\tau}(W_t^{\pi}, \pi_t) - W_t^{\pi} \le L \mid \mathcal{F}_t\right) < \alpha \right\}$$

is the quantile of the projected portfolio gain over the interval $(t, t+\tau)$ and $x^- = \max[0, -x]$. In other words, $VaR_t^{\alpha,\pi}$ is the loss over the next period of length τ which would be exceeded only with a (small) conditional probability α if the current portfolio π_t were kept unchanged.

¹³If n > d or rank $(\sigma) < n$, some stocks are redundant and can be omitted from the analysis.

¹⁴All the inequalities involving random variables are understood to hold almost surely.

The fact that $VaR_t^{\alpha,\pi}$ is computed under the assumption that the current portfolio is kept unchanged reflects the actual practice and the fact that financial institutions monitoring their traders do not typically know the traders' future portfolio choices over the VaR horizon. Instead, the measure of VaR in (4) only requires knowledge of the current portfolio value, the current portfolio composition and the conditional distribution of asset returns.¹⁵

Similarly, the TCE of a portfolio $\pi \in \Pi$ is defined by

$$TCE_t^{\alpha,\pi} = \left(\frac{\mathrm{E}\left[(W_t^{\pi} - \mathcal{W}_{t+\tau}(W_t^{\pi}, \pi_t))\mathbf{1}_{\{W_t^{\pi} - \mathcal{W}_{t+\tau}(W_t^{\pi}, \pi_t) \ge -Q_t^{\alpha,\pi}\}} \mid \mathcal{F}_t\right]}{\alpha}\right)^+, \qquad (5)$$

where $x^+ = \max[0, x]$. In other words, the TCE of a portfolio is the conditional expected value of the losses exceeding $-Q_t^{\alpha, \pi}$.

Given our assumption of lognormally-distributed asset returns, both the VaR and the TCE of a portfolio can be explicitly computed.

Proposition 1. We have

$$VaR_t^{\alpha,\pi} = W_t^{\pi} \left[1 - \exp\left(\left(r + \pi_t^{\mathsf{T}} \mu - \frac{1}{2} |\pi_t^{\mathsf{T}} \sigma|^2 \right) \tau + N^{-1}(\alpha) |\pi_t^{\mathsf{T}} \sigma| \sqrt{\tau} \right) \right]^+ \tag{6}$$

and

$$TCE_t^{\alpha,\pi} = W_t^{\pi} \left[1 - \exp\left(\left(r + \pi_t^{\mathsf{T}} \mu\right) \tau\right) \frac{N\left(N^{-1}(\alpha) - |\pi_t^{\mathsf{T}} \sigma| \sqrt{\tau}\right)}{\alpha} \right]^+, \tag{7}$$

where N(x) and $N^{-1}(x)$ denote the normal distribution and inverse distribution functions. In particular,

$$0 \le VaR_t^{\alpha,\pi} \le TCE_t^{\alpha,\pi} < W_t^{\pi} \tag{8}$$

and

$$VaR_t^{\alpha,0} = TCE_t^{\alpha,0} = 0.$$
(9)

PROOF. See the Appendix.

3. Optimal Trading Strategies under VaR Limits

Now consider the problem of a trader who starts with an endowment W_0 and must select a portfolio $\pi \in \Pi$ so as to maximize the expected utility $\mathbf{E}[u(W_T^{\pi})]$ of the terminal value of

¹⁵Alternatively, it would be possible to compute the portfolio's VaR under the assumptions that the current asset holdings were kept unchanged. Our formulation not only has computational advantages under lognormality, but is also a natural one for portfolios having target compositions specified in terms of proportions rather than amounts. In addition, this formulation is consistent with typical calculations of the VaR of a portfolio, which are based on the assumption that the horizon portfolio return equals a weighted average of the horizon asset returns, with weights equals to the relative amounts invested at the beginning of the period (see Jorion (2001, Section 7.1)). In any case, the difference between the two formulations is likely to be insignificant if the VaR horizon τ is small (e.g., 1 day).

the trading portfolio, subject to the constraint that, at any time $t \in [0, T]$, the value at risk of its portfolio, $VaR_t^{\alpha,\pi}$, is no larger than some prespecified level $VaR(W_t^{\pi}, t) \geq 0$:

s.t.
$$\begin{aligned}
\max_{\pi \in \Pi} & \mathbb{E}[u(W_T^{\pi})] \\
W_0^{\pi} &= W_0 \\
& VaR_t^{\alpha,\pi} \leq \overline{VaR}(W_t^{\pi}, t) \quad \forall t \in [0, T].
\end{aligned} \tag{10}$$

Note that in (10) we allow the VaR limit at time t to depend on calendar time and on the current value of the portfolio.

Remark 1. Since we have assumed $\overline{VaR}(W_t^{\pi}, t) \ge 0$, it follows from (9) that setting $\pi_t = 0$ (that is, investing everything in the riskless bond) always satisfies the VaR constraint. Hence, the set of feasible trading strategies is not empty.

Remark 2. The expression for VaR in (6) implies that a portfolio π satisfies the constraint $VaR_t^{\alpha,\pi} \leq \overline{VaR}(W_t^{\pi},t)$ if and only if

$$\log\left(1 - \frac{\overline{VaR}(W_t^{\pi}, t)}{W_t^{\pi}}\right)^+ - \left(r + \pi_t^{\mathsf{T}}\mu - \frac{1}{2}|\pi_t^{\mathsf{T}}\sigma|^2\right)\tau - N^{-1}(\alpha)|\pi_t^{\mathsf{T}}\sigma|\sqrt{\tau} \le 0.$$
(11)

In the case of a single risky asset (n = 1), it can be easily verified that (11) is equivalent to an upper and a lower bound on the fraction π_t allocated to the risky asset:

$$\pi^{-}(W_{t}^{\pi},t) \le \pi_{t} \le \pi^{+}(W_{t}^{\pi},t),$$

where

$$\pi^{\pm}(W,t) = \frac{\frac{\mu}{|\sigma|}\sqrt{\tau} \pm N^{-1}(\alpha) \pm \sqrt{\left(\frac{\mu}{|\sigma|}\sqrt{\tau} \pm N^{-1}(\alpha)\right)^2 - 2\left(\log\left(1 - \frac{\overline{VaR}(W,t)}{W}\right)^+ - r\tau\right)}}{|\sigma|\sqrt{\tau}}.$$
(12)

In particular, given the current portfolio value W_t^{π} , the set of admissible portfolios π_t is convex. This is however generally not the case in the presence of multiple risky assets if $\alpha > \frac{1}{2}$. Figure 1 shows an example with two risky assets in which the set of admissible trading strategies is not convex.¹⁶

Rewriting (10) as the stochastic control problem

s.t.
$$\begin{aligned} \max_{\pi \in \Pi} \mathbf{E}[u(W_T^{\pi})] \\ & \left[W_t^{\pi} = W_0 + \int_0^t W_s^{\pi} \left(r + \pi_s^{\top} \mu \right) \, ds + \int_0^t W_s^{\pi} \pi_s^{\top} \sigma \, dw_s, \\ & \left[\log \left(1 - \frac{\overline{VaR}(W_t^{\pi}, t)}{W_t^{\pi}} \right)^+ - \left(r + \pi_t^{\top} \mu - \frac{1}{2} |\pi_t^{\top} \sigma|^2 \right) \tau - N^{-1}(\alpha) |\pi_t^{\top} \sigma| \sqrt{\tau} \le 0 \end{aligned}$$
(13)

leads to the following characterization of optimal trading strategies.¹⁷

¹⁶Note that in this example both assets have a return risk premium of .07 and a volatility of .17, while the instantaneous correlation coefficient between the two risky assets is -.9.

¹⁷Since the set of π_t satisfying the inequality (11) is not necessarily convex when n > 1 (as shown in Figure 1) and it depends on the current portfolio value W_t^{π} , the convex duality technique of Cvitanić and Karatzas (1992) cannot be applied to this problem.



Figure 1: The graph shows the set of portfolios satisfying the VaR constraint, assuming n = 2, r = .008, $\mu = \begin{pmatrix} .07 \\ .07 \end{pmatrix}$, $\sigma = \begin{pmatrix} .170 & 0 \\ -.153 & .074 \end{pmatrix}$, $\overline{\frac{VaR(W_t^{\pi}, t)}{W_t^{\pi}}} = .01$, $\alpha = .9$, $\tau = 1$.

Theorem 1. Let V(W,t) denote the value function for the stochastic control problem (13) and let

$$\varphi_{\alpha}^{+}(W,t) = \frac{|\kappa|\sqrt{\tau} + N^{-1}(\alpha) + \sqrt{(|\kappa|\sqrt{\tau} + N^{-1}(\alpha))^2 - 2\left(\log\left(1 - \frac{\overline{VaR}(W,t)}{W}\right)^+ - r\tau\right)}}{|\kappa|\sqrt{\tau}},$$
(14)

where $\kappa = \sigma^{\top}(\sigma\sigma^{\top})^{-1}\mu$. Then $\varphi_{\alpha}^{+}(W,t) \geq 0$ for all $(W,t) \in (0,\infty) \times [0,T]$ and V solves the Hamilton-Jacobi-Bellman equation

$$0 = \begin{cases} -\frac{1}{2} \frac{V_W^2}{V_{WW}} |\kappa|^2 + V_W W r + V_t & \text{if } -\frac{V_W}{WV_{WW}} \le \varphi_\alpha^+ \\ \frac{1}{2} V_{WW} W^2 |\kappa \varphi_\alpha^+|^2 + V_W W (r + |\kappa|^2 \varphi_\alpha^+) + V_t & \text{otherwise} \end{cases}$$
(15)

with terminal condition

$$V(W,T) = u(W). (16)$$

Finally, letting

$$\varphi(W,t) = \min\left[-\frac{V_W(W,t)}{WV_{WW}(W,t)}, \varphi^+_{\alpha}(W,t)\right]$$
(17)

the policy

$$\pi^*(W,t) = \varphi(W,t)(\sigma\sigma^{\top})^{-1}\mu$$
(18)

solves (13).

PROOF. See the Appendix.

Remark 3. Equation (18) shows that the optimal portfolio of a VaR-constrained agent is a combination of the riskless asset and the growth-optimal portfolio $(\sigma\sigma^{\top})^{-1}\mu$.¹⁸ Thus, with lognormally-distributed asset returns, the VaR constraint affects the distribution of the optimal portfolio between riskless and risky assets, but does not distort the composition of the optimal portfolio of risky assets. The function $\varphi^+_{\alpha}(W,t)$ in (14) identifies the maximum fraction of wealth that can be invested in the growth-optimal portfolio at time t under the VaR constraint¹⁹ (as shown in the proof of Theorem 1, shorting the growth-optimal portfolio is never optimal).

The result in Theorem 1 also allows us to compute the distribution of the terminal portfolio value under the optimal trading strategy.

Corollary 1. Let p(W,t) denote the density function of $W_t^{\pi^*}$. Then p solves Kolmogorov's forward equation

$$\frac{\partial}{\partial t}p = \frac{\partial^2}{\partial W^2} \left[(W\varphi|\kappa|)^2 p \right] - \frac{\partial}{\partial W} \left[W(r+\varphi|\kappa|^2) p \right] (Wp)$$
(19)

with initial condition

 $p(W,0) = \delta(W - W_0),$

where φ is the function in (17) and δ denotes Dirac's delta function.

PROOF. See Karatzas and Shreve (1988).

4. Examples with CRRA Utility

We now specialize our model by assuming that $u(W) = \frac{W^{1-\gamma}}{1-\gamma}$ for some $\gamma > 0$. We recall that, in the absence of a VaR constraint,

$$V(W,t) = e^{\rho(T-t)} \frac{W^{1-\gamma}}{1-\gamma}$$

where

$$\rho = (1 - \gamma) \left(r + \frac{|\kappa|^2}{2\gamma} \right),$$

and

$$\pi^*(W,t) = \frac{1}{\gamma} (\sigma \sigma^{\top})^{-1} \mu.$$
(20)

Thus, it follows from (2) and (20) that the terminal portfolio value $W_T^{\pi^*}$ is in this case lognormally distributed, with mean

$$W_0 e^{\left(r + \frac{|\kappa|^2}{\gamma}\right)T}$$

¹⁸The growth-optimal portfolio is the portfolio that maximizes the expected continuously-compounded rate of return $\frac{1}{T} \log (W_T^{\pi}/W_0^{\pi})$. Equivalently, π^* is a combination of the riskless asset and the mean-variance efficient portfolio of risky assets $(\sigma\sigma^{\top})^{-1}\mu/\bar{1}^{\top}(\sigma\sigma^{\top})^{-1}\mu$.

¹⁹This can be immediately verified by noting that the expression for φ_{α}^{+} in (14) can be obtained from the expression for π^{+} in (12) by replacing μ with $|\kappa|^{2} = \mu^{\top} (\sigma \sigma^{\top})^{-1} \mu$ (the instantaneous risk premium on the growth-optimal portfolio) and $|\sigma|$ with $|\kappa| = |\mu^{\top} (\sigma \sigma^{\top})^{-1} \sigma|$ (the volatility of the growth-optimal portfolio).

and standard deviation

$$W_0 e^{\left(r + \frac{|\kappa|^2}{\gamma}\right)T} \sqrt{e^{\frac{|\kappa|^2}{\gamma}T} - 1}.$$

To further understand the implications of VaR constraints for optimal trading strategies, we consider below three alternative specifications of the function $\overline{VaR}(W,t)$ which identifies the maximum admissible VaR at any time $t \in [0, T]$. Notice that it follows immediately from Remark 3 and (20) that a given VaR constraint not binding if and only if

$$\frac{1}{\gamma} \le \inf_{(W,t)\in(0,\infty)\times[0,T]} \varphi_{\alpha}^+(W,t).$$

Moreover, it follows from (18) and (20) that the VaR-constrained optimal portfolio is a multiple

$$q(W,t) = \gamma \varphi(W,t) \tag{21}$$

of the unconstrained optimal portfolio. Following the terminology of Basak and Shapiro (2001), we will refer to this multiple as the *relative risk exposure*. In particular, we have from the boundary condition (16) that at the terminal date

$$q(W,T) = \min[1, \gamma \varphi_{\alpha}^{+}(W,T)].$$

In cases where an analytical solution is not available, we solve the PDE (15) numerically by rewriting it in terms of the state variable $w = \log(W)$ and then applying the explicit finite-difference method (see Kushner and Dupuis (1992) for details). This allows us to obtain the optimal trading strategy π^* from (18). Since the finite-difference method approximates the state variable W^{π^*} with a Markov chain with known transition probabilities, we use these transition probabilities to compute the distribution of the terminal portfolio value $W_T^{\pi^*}$. This approach generates results similar to those obtainable by solving the Kolmogorov equation (19) separately using the finite-difference method.

All the numerical computations assume r = .008, $|\kappa| = .37$, $\alpha = .01$, $\tau = 1/260$ (one trading day), T = 10, $\gamma = .5$ or $\gamma = 5$ and $W_0 = 1$. Moreover, we scale the function $\overline{VaR}(W,t)$ so that $\overline{VaR}(W_0,0) = .05$ (this amounts to setting $\beta = .05$ in the examples below).

4.1. $\overline{VaR}(W,t) = \beta$

We start by considering the case of a constant VaR limit: $\overline{VaR}(W,t) = \beta$. In this case, the function φ_{α}^+ is independent of t and monotonically decreasing in W, with

$$\lim_{W\downarrow 0}\varphi_{\alpha}^{+}(W,t)=+\infty$$

and

$$\lim_{W\uparrow+\infty}\varphi_{\alpha}^{+}(W,t) = \frac{|\kappa|\sqrt{\tau} + N^{-1}(\alpha) + \sqrt{(|\kappa|\sqrt{\tau} + N^{-1}(\alpha))^{2} + 2r\tau}}{|\kappa|\sqrt{\tau}} = .000582.$$

Therefore, the VaR constraint is binding if $\gamma < \frac{1}{000582} = 1,717.8$ and not binding otherwise.



Figure 2: The graph plots the relative risk exposure q(W,t) at t = 0 (heavier line) and t = T (lighter line), assuming r = .008, $|\kappa| = .37$, T = 10, $\gamma = .5$, $W_0 = 1$, $\alpha = .01$, $\tau = 1/260$, $\overline{VaR}(W,t) = .05$.



Figure 3: The graph plots the probability density of the terminal portfolio value $W_T^{\pi^*}$ under the optimal trading strategy in the constrained (heavier line) and unconstrained (lighter line) case, assuming r = .008, $|\kappa| = .37$, T = 10, $\gamma = .5$, $W_0 = 1$, $\alpha = .01$, $\tau = 1/260$, $\overline{VaR}(W, t) = .05$.

Figure 2 plots the relative risk exposure under the optimal policy (the function q(W, t) in (21)) for the case $\gamma = .5$, when t = 0 and when t = T. Contrary to the conclusion of Basak and Shapiro (2001), the figure shows that a VaR-constrained agent never invests more in risky assets than a VaR-unconstrained agent (the relative risk exposure is never larger than 1). Consequently, as shown in Figure 3, the probability of extreme losses at the horizon T is lower under the VaR-constrained investment strategy than under the unconstrained strategy. These conclusions also apply to the other examples we consider. Thus, the reservations expressed by Basak and Shapiro (2001), Vorst (2001) and others against the use of VaR as a risk-control tool seem unwarranted if the VaR is reevaluated periodically.

It is also worth noting the presence of a significant hedging demand in Figure 2: for example, a VaR-constrained agent with initial wealth $W_0 = .4$ (log(W_0) = -.916) and an investment horizon of 10 years would invest only 67% as much as an unconstrained agent in the growth-optimal portfolio, even though he could invest the unconstrained-optimal amount in the growth-optimal portfolio and still satisfy the VaR constraint at time 0 (since $\varphi^+_{\alpha}(.4, 0) = 2.5$). Clearly, this lower allocation to the growth-optimal portfolio reduces the volatility of the optimal portfolio and reflects the smaller indirect utility of extreme portfolio values induced by the fact that a constant VaR constraint becomes more severely binding when the portfolio value increases (as can be seen from Figure 2 or from the fact that the function φ^+ in (14) is a decreasing function of the portfolio value in this case).

Figures 4 and 5 illustrate the corresponding results for the case $\gamma = 5$. In this case, hedging demand is negligible, as shown by the fact that the optimal risk exposure at the initial date is very close to that at the terminal date. This stems from the fact that the VaR constraint is binding at time t (i.e., $\frac{1}{\gamma} > \varphi_{\alpha}^{+}(W, t)$) only if W > 4.76 (log(W) > 1.56), and this event has negligible probability, as shown in Figure 5.

4.2. $\overline{VaR}(W,t) = \beta W$

Fixing the VaR limit to a constant amount has the obvious shortcoming that the constraint becomes binding when the portfolio value increases and is not binding when the portfolio value is sufficiently low. Thus, a constant VaR limit penalizes successful traders. In practice, successful traders typically see their VaR limit increased. To capture this fact, we consider next the case of a constant proportional VaR, $\overline{VaR}(W, t) = \beta W$.²⁰

Thus,

$$\varphi_{\alpha}^{+}(W,t) = \frac{|\kappa|\sqrt{\tau} + N^{-1}(\alpha) + \sqrt{(|\kappa|\sqrt{\tau} + N^{-1}(\alpha))^{2} - 2\left(\log(1-\beta)^{+} - r\tau\right)}}{|\kappa|\sqrt{\tau}} = \varphi_{\alpha}^{+} \quad (22)$$

for all (W, t). It can be easily verified that in this case the value function

$$V(W,t) = e^{\hat{\rho}(T-t)} \frac{W^{1-\gamma}}{1-\gamma},$$

²⁰Alternatively, the proportional constraint $VaR_t^{\alpha,\pi} \leq \beta W_t^{\pi}$ (or. equivalently, $W_t^{\pi} \geq \beta^{-1} VaR_t^{\alpha,\pi}$) could be motivated by the requirement that capital exceed a given multiple β^{-1} of the current VaR, as advocated by the Basle Committee proposals of April 1995.



Figure 4: The graph plots the relative risk exposure q(W,t) at t = 0 (heavier line) and t = T (lighter line), assuming r = .008, $|\kappa| = .37$, T = 10, $\gamma = 5$, $W_0 = 1$, $\alpha = .01$, $\tau = 1/260$, VaR(W,t) = .05.



Figure 5: The graph plots the probability density of the terminal portfolio value $W_T^{\pi^*}$ under the optimal trading strategy in the constrained (heavier line) and unconstrained (lighter line) case, assuming r = .008, $|\kappa| = .37$, T = 10, $\gamma = 5$, $W_0 = 1$, $\alpha = .01$, $\tau = 1/260$, $\overline{VaR}(W, t) = .05$.



Figure 6: The graph plots the probability density of the terminal portfolio value $W_T^{\pi^*}$ under the optimal trading strategy in the constrained (heavier line) and unconstrained (lighter line) case, assuming r = .008, $|\kappa| = .37$, T = 10, $\gamma = .5$, $W_0 = 1$, $\alpha = .01$, $\tau = 1/260$, $\overline{VaR}(W, t) = .05W$.

where

$$\hat{\rho} = (1 - \gamma) \left(r + \varphi^* \left(1 - \frac{\gamma \varphi^*}{2} \right) |\kappa|^2 \right)$$

and

$$\varphi^* = \min\left(\frac{1}{\gamma}, \varphi^+_\alpha\right),$$

solves the HJB equation (15). Thus, $\varphi(W,t) = \varphi^*$ and $q(W,t) = \gamma \varphi^* \leq 1$ for all (W,t). Hence, in this case there is no hedging demand and the VaR-constrained optimal trading strategy coincides with the unconstrained-optimal trading strategy for an investor with CRRA coefficient $\gamma^* = \frac{1}{\varphi^*}$.

With the parameters of our numerical example, $\varphi_{\alpha}^{+} = .966$, so that the VaR constraint is binding if $\gamma < \frac{1}{.966} = 1.035$ and not binding otherwise. In particular, if $\gamma = .5$ the constrained-optimal trading strategy coincides with the unconstrained trading strategy for an agent with higher CRRA coefficient $\gamma^* = 1.035$ and the relative risk exposure q(W,t) is constant and equal to .483. Figure 6 shows the distribution of the terminal portfolio value under the VaR-constrained and the unconstrained optimal trading strategies in this case.

4.3. $\overline{VaR}(W,t) = (W - (1 - \beta)W_0)^+$

In the example of Section 4.1. (constant VaR) the maximum allowable proportional investment in the growth-optimal portfolio, φ_{α}^{+} , was a decreasing function of the current portfolio value, while in the example of Section 4.2. (constant proportional VaR) it was a constant. As a final example, we consider the case in which $\overline{VaR}(W,t) = (W - (1 - \beta)W_0)^+$: thus, the VaR limit equals a fixed proportion βW_0 of the initial portfolio value, plus any running gain $W - W_0$. In this case, $\varphi_{\alpha}^+(W,t)$ is a monotonically increasing function of W, with

$$\lim_{W \downarrow 0} \varphi_{\alpha}^{+}(W,t) = \frac{|\kappa|\sqrt{\tau} + N^{-1}(\alpha) + \sqrt{(|\kappa|\sqrt{\tau} + N^{-1}(\alpha))^{2} + 2r\tau}}{|\kappa|\sqrt{\tau}} = .000582$$

and

$$\lim_{W\uparrow+\infty}\varphi_{\alpha}^{+}(W,t)=+\infty.$$

Therefore, the VaR constraint is binding if $\gamma < \frac{1}{.000582} = 1,717.8$ and not binding otherwise (as in the example of Section 4.1.).

It is also worth noting that, in the extreme case $\alpha = 0$, $\varphi_{\alpha}^{+}(W,t) = 0$ for all $W \leq (1-\beta)W_0$, so that the optimal portfolio π^* has the property that $W_t^{\pi^*} \geq (1-\beta)W_0$ for all $t \in [0,T]$. Thus, the dynamic VaR constraint $VaR_t^{\alpha,\pi} \leq (W_t^{\pi} - (1-\beta)W_0)^+$ can be considered as a relaxed version of the dynamic portfolio insurance constraint $W_t^{\pi} \geq (1-\beta)W_0$ for all $t \in [0,T]$.

Figures 7 and 8 (respectively, Figures 9 and 10) show the optimal risk exposure and the distribution of the terminal portfolio value when $\gamma = .5$ (respectively, $\gamma = 5$). In both cases, the VaR constraint results in a highly skewed distribution for the terminal portfolio value. As already noted, when $\alpha = 0$ the resulting distribution must assign zero probability to values of $\log(W_T^{\pi^*})$ below $\log[(1 - \beta)W_0] = -.051$: in Figures 8 and 10 the probability of these values is positive but negligible, while the probability of a loss smaller than βW_0 is significantly larger than under the unconstrained policy. For the larger value of the risk aversion coefficient, the risk limit does not bind if the portfolio value is above the initial value and the probability of a large terminal gain is close to that in the unconstrained case.

5. TCE Limits

We now turn to the problem of a trader subject to a risk limit specified in terms of TCE,

s.t.
$$\max_{\substack{\pi \in \Pi \\ \pi \in \Pi}} E[u(W_T^{\pi})]$$
$$W_0^{\pi} = W_0$$
$$TCE_t^{\hat{\alpha},\pi} \leq \overline{TCE}(W_t^{\pi},t) \quad \forall t \in [0,T],$$
(23)

where \overline{TCE} is a given nonnegative function and $\hat{\alpha} \in (0, 1)$. As mentioned in the Introduction, TCE has been advocated as a better risk management tool than VaR. As we will see, however, any dynamic risk limit formulated in terms of TCE can be easily mapped into an equivalent VaR limit, and conversely, so that the choice of VaR or TCE as a riskmanagement tool is largely irrelevant.

Definition. The constraints $VaR_t^{\alpha,\pi} \leq \overline{VaR}(W_t^{\pi},t)$ and $TCE_t^{\hat{\alpha},\pi} \leq \overline{TCE}(W_t^{\pi},t)$ are equivalent if the optimal portfolio policies in (10) and (23) coincide for all utility functions u.

Recall from Proposition 1 that

$$VaR_t^{\alpha,\pi} = W_t^{\pi}\rho_{\alpha}(\pi_t)^+$$



Figure 7: The graph plots the relative risk exposure q(W,t) at t = 0 (heavier line) and t = T (lighter line), assuming r = .008, $|\kappa| = .37$, T = 10, $\gamma = .5$, $W_0 = 1$, $\alpha = .01$, $\tau = 1/260$, $\overline{VaR}(W,t) = (W - (1 - .05)W_0)^+$.



Figure 8: The graph plots the probability density of the terminal portfolio value $W_T^{\pi^*}$ under the optimal trading strategy in the constrained (heavier line) and unconstrained (lighter line) case, assuming r = .008, $|\kappa| = .37$, T = 10, $\gamma = .5$, $W_0 = 1$, $\alpha = .01$, $\tau = 1/260$, $\overline{VaR}(W,t) = (W - (1 - .05)W_0)^+$.



Figure 9: The graph plots the relative risk exposure q(W,t) at t = 0 (heavier line) and t = T (lighter line), assuming r = .008, $|\kappa| = .37$, T = 10, $\gamma = 5$, $W_0 = 1$, $\alpha = .01$, $\tau = 1/260$, $\overline{VaR}(W,t) = (W - (1 - .05)W_0)^+$.



Figure 10: The graph plots the probability density of the terminal portfolio value $W_T^{\pi^*}$ under the optimal trading strategy in the constrained (heavier line) and unconstrained (lighter line) case, assuming r = .008, $|\kappa| = .37$, T = 10, $\gamma = 5$, $W_0 = 1$, $\alpha = .01$, $\tau = 1/260$, $\overline{VaR}(W, t) = (W - (1 - .05)W_0)^+$.

and

$$TCE_t^{\hat{\alpha},\pi} = W_t^{\pi}\hat{\rho}_{\hat{\alpha}}(\pi_t)^+,$$

where

$$\rho_{\alpha}(\pi_t) = 1 - \exp\left(\left(r + \pi_t^{\top} \mu - \frac{1}{2} |\pi_t^{\top} \sigma|^2\right) \tau + N^{-1}(\alpha) |\pi_t^{\top} \sigma| \sqrt{\tau}\right)$$

and

$$\hat{\rho}_{\hat{\alpha}}(\pi_t) = 1 - \exp\left(\left(r + \pi_t^\top \mu\right)\tau\right) \frac{N\left(N^{-1}(\hat{\alpha}) - |\pi_t^\top \sigma|\sqrt{\tau}\right)}{\hat{\alpha}}.$$

Since $TCE_t^{\alpha,\pi}$ depends on π only through the instantaneous expected rate of return $\pi_t^{\top}\mu$ and the instantaneous return variance $|\pi_t^{\top}\sigma|^2$, it follows as in Theorem 1 that the optimal portfolio π^* in (23) is a combination of the riskless asset and the growth-optimal portfolio $(\sigma\sigma^{\top})^{-1}\mu$, i.e.,

$$\pi_t^* = \hat{\varphi}(W_t^{\pi^*}, t) (\sigma \sigma^{\top})^{-1} \mu \tag{24}$$

for some function $\hat{\varphi}$. The next lemma implies that, for portfolios that are combinations of the riskless asset and the growth-optimal portfolio, the TCE limit is equivalent to a lower and an upper bound on the allocation to the growth-optimal portfolio.

Lemma 1. For all $a \ge 0$, the set

$$\Phi^{a}_{\hat{\alpha}} = \left\{ \varphi \in I\!\!R : \hat{\rho}_{\hat{\alpha}}(\varphi(\sigma\sigma^{\top})^{-1}\mu) \leq a \right\}$$

is a closed interval containing the origin. Thus, there exist functions $\Phi_{\hat{\alpha}}^-$ and $\Phi_{\hat{\alpha}}^+$ with $\Phi_{\hat{\alpha}}^- < 0 < \Phi_{\hat{\alpha}}^+$ such that $\Phi_{\hat{\alpha}}^a = [\Phi_{\hat{\alpha}}^-(a), \Phi_{\hat{\alpha}}^+(a)]$. If a < 1, $\Phi_{\hat{\alpha}}^a$ is bounded and $\Phi_{\hat{\alpha}}^-(a)$ and $\Phi_{\hat{\alpha}}^+(a)$ are the two roots of the equation $\hat{\rho}_{\hat{\alpha}}(\varphi(\sigma\sigma^{\top})^{-1}\mu) = a$. If $a \ge 1$, $\Phi_{\hat{\alpha}}^-(a) = -\infty$ and $\Phi_{\hat{\alpha}}^+(a) = +\infty$.

PROOF. See the Appendix.

Recalling that $TCE_t^{\hat{\alpha},\pi} = W_t^{\pi}\hat{\rho}_{\hat{\alpha}}(\pi_t)^+$, it follows immediately from the above lemma that the policy π^* in (24) satisfies the TCE limit in (23) if and only if

$$\hat{\varphi}_{\hat{\alpha}}^{-}(W,t) \leq \hat{\varphi}(W,t) \leq \hat{\varphi}_{\hat{\alpha}}^{+}(W,t)$$

for all $(W, t) \in \mathbb{R}^+ \times [0, T]$, where

$$\hat{\varphi}^{\pm}_{\hat{\alpha}}(W,t) = \Phi^{\pm}_{\hat{\alpha}}\left(\frac{\overline{TCE}(W,t)}{W}\right).$$
(25)

Since shorting the growth-optimal portfolio is never optimal,²¹ it is then easy to show that given VaR and TCE limits are equivalent if and only if the maximum feasible allocation to the growth-optimal portfolios under the two constraints coincide, i.e., if the function φ_{α}^{+} in (14) coincides with the function $\hat{\varphi}_{\hat{\alpha}}^{+}$ in (25). This leads to the following result.

²¹This follows from the same argument used in the proof of Theorem 1.

Proposition 2. The constraint $TCE_t^{\hat{\alpha},\pi} \leq \overline{TCE}(W_t^{\pi},t)$ is equivalent to the constraint $VaR_t^{\alpha,\pi} \leq \overline{VaR}(W_t^{\pi},t)$, where $\alpha \in (0,1)$ is an arbitrary probability such that

$$\frac{|\kappa|\sqrt{\tau} + N^{-1}(\alpha) + \sqrt{(|\kappa|\sqrt{\tau} + N^{-1}(\alpha))^2 + 2r\tau}}{|\kappa|\sqrt{\tau}} \le \inf_{(W,t)\in\mathbb{R}^+\times[0,T]}\hat{\varphi}^+_{\hat{\alpha}}(W,t), \quad (26)$$

$$\overline{VaR}(W,t) = W\rho_{\alpha}(\hat{\varphi}^{+}_{\hat{\alpha}}(W,t)(\sigma\sigma^{\top})^{-1}\mu) \ge 0, \qquad (27)$$

and $\varphi_{\hat{\alpha}}^+$ is the function in (25). Conversely, the constraint $VaR_t^{\alpha,\pi} \leq \overline{VaR}(W_t^{\pi},t)$ is equivalent to the constraint $TCE_t^{\hat{\alpha},\pi} \leq \overline{TCE}(W_t^{\pi},t)$, where $\hat{\alpha} \in (0,1)$ is an arbitrary probability such that

$$\Phi_{\hat{\alpha}}^{+}(0) \leq \inf_{(W,t)\in\mathbb{R}^{+}\times[0,T]}\varphi_{\alpha}^{+}(W,t),$$
(28)

$$\overline{TCE}(W,t) = W\hat{\rho}_{\hat{\alpha}}(\varphi_{\alpha}^{+}(W,t)(\sigma\sigma^{\top})^{-1}\mu) \ge 0,$$
(29)

and φ_{α}^{+} is the function in (14)

PROOF. See the Appendix.

Proposition 2 implies in particular that a proportional VaR limit is equivalent to a proportional TCE limit.

Corollary 2. A proportional VaR limit $VaR_t^{\alpha,\pi} \leq \beta W$ with $\beta \in (0,1)$ is equivalent to a proportional TCE limit $TCE_t^{\alpha,\pi} \leq \hat{\beta}W$, where

$$\beta \leq \hat{\beta} = \hat{\rho}_{\alpha} \left(\frac{|\kappa|\sqrt{\tau} + N^{-1}(\alpha) + \sqrt{(|\kappa|\sqrt{\tau} + N^{-1}(\alpha))^2 - 2\left(\log(1-\beta) - r\tau\right)}}{|\kappa|\sqrt{\tau}} (\sigma\sigma^{\top})^{-1} \mu \right) < 1$$

PROOF. See the Appendix.

6. Concluding Remarks

A frequently-mentioned limitation of VaR as a risk-control tool is that VaR focuses on the *probability* of large losses, but not on the *expected value* of these losses. This might induce traders subject to VaR limits to post large losses in the exceptional cases where losses exceed the VaR limit and has led several authors to propose alternatives to VaR based on the expected value of large losses. In this paper we show that this intuition, largely developed from static models, does not apply to dynamic models where the VaR is reevaluated periodically, making full use of conditioning information. Instead, in all the cases we consider, we always find that the expected value of losses and the proportional investment in risky assets are lower under a VaR constraint than they would have been without the constraint. In addition, we show that, in spite of the fact that VaR in not a coherent risk measure, risk limits formulated in terms of VaR are equivalent to to risk limits formulated in terms of TCE, which is known to be a coherent risk measure. These findings provide some theoretical support for the growing use of VaR as a risk-control tool. It remains to be seen if and to what extent they apply to models with varying price coefficients.

Appendix

PROOF OF PROPOSITION 1: We have

$$P\left(\mathcal{W}_{t+\tau}(W_t^{\pi}, \pi_t) - W_t^{\pi} \le L \mid \mathcal{F}_t\right)$$

$$= P\left(\exp\left(\left(r + \pi_t^{\top} \mu - \frac{1}{2}|\pi_t^{\top} \sigma|^2\right)\tau + \pi_t^{\top} \sigma(w_{t+\tau} - w_t)\right) \le 1 + \frac{L}{W_t^{\pi}} \mid \mathcal{F}_t\right)$$

$$= P\left(\pi_t^{\top} \sigma(w_{t+\tau} - w_t) \le \log\left(1 + \frac{L}{W_t^{\pi}}\right)^+ - \left(r + \pi_t^{\top} \mu - \frac{1}{2}|\pi_t^{\top} \sigma|^2\right)\tau \mid \mathcal{F}_t\right)$$

$$= N\left(\frac{\log\left(1 + \frac{L}{W_t^{\pi}}\right)^+ - \left(r + \pi_t^{\top} \mu - \frac{1}{2}|\pi_t^{\top} \sigma|^2\right)\tau}{|\pi_t^{\top} \sigma|\sqrt{\tau}}\right),$$

where the last equality follows from the fact that the random variable $\pi_t^{\top} \sigma(w_{t+\tau} - w_t)$ is conditionally normally distributed with zero mean and variance $|\pi_t^{\top} \sigma|^2 \tau$. Thus,

$$P\left(\mathcal{W}_{t+\tau}(W_t^{\pi}, \pi_t) - W_t^{\pi} \le L \mid \mathcal{F}_t\right) \le \alpha$$

$$\iff N\left(\frac{\log\left(1 + \frac{L}{W_t^{\pi}}\right)^+ - \left(r + \pi_t^{\top}\mu - \frac{1}{2}|\pi_t^{\top}\sigma|^2\right)\tau}{|\pi_t^{\top}\sigma|\sqrt{\tau}}\right) \le \alpha$$

$$\iff L \le W_t^{\pi}\left[\exp\left(\left(r + \pi_t^{\top}\mu - \frac{1}{2}|\pi_t^{\top}\sigma|^2\right)\tau + N^{-1}(\alpha)|\pi_t^{\top}\sigma|\sqrt{\tau}\right) - 1\right],$$

which implies

$$Q_t^{\alpha,\pi} = W_t^{\pi} \left[\exp\left(\left(r + \pi_t^{\mathsf{T}} \mu - \frac{1}{2} | \pi_t^{\mathsf{T}} \sigma |^2 \right) \tau + N^{-1}(\alpha) | \pi_t^{\mathsf{T}} \sigma | \sqrt{\tau} \right) - 1 \right].$$

Therefore,

$$VaR_t^{\alpha,\pi} = (Q_t^{\alpha,\pi})^- = W_t^{\pi} \left[1 - \exp\left(\left(r + \pi_t^{\top} \mu - \frac{1}{2} |\pi_t^{\top} \sigma|^2\right) \tau + N^{-1}(\alpha) |\pi_t^{\top} \sigma| \sqrt{\tau}\right) \right]^+.$$

Similarly,

Dividing by α gives the expression for $TCE_t^{\alpha,\pi}$ in (7).

PROOF OF THEOREM 1: The Hamilton-Jacobi-Bellman (HJB) equation for the problem in (13) is

$$0 = \max_{\pi} \left[\frac{1}{2} V_{WW} | W \pi^{\top} \sigma |^{2} + V_{W} W (r + \pi^{\top} \mu) + V_{t} - \psi \left(\log \left(1 - \frac{\overline{VaR}}{W} \right)^{+} - \left(r + \pi^{\top} \mu - \frac{1}{2} | \pi^{\top} \sigma |^{2} \right) \tau - N^{-1}(\alpha) | \pi^{\top} \sigma | \sqrt{\tau} \right) \right],$$
(30)

where ψ is a Lagrangian multiplier. The first-order conditions for a maximum in (30) are:

$$V_{WW}W^2\sigma\sigma^{\top}\pi^* + V_WW\mu + \psi\left((\mu - \sigma\sigma^{\top}\pi^*)\tau + N^{-1}(\alpha)\frac{\sigma\sigma^{\top}\pi^*}{|\pi^{*\top}\sigma|}\sqrt{\tau}\right) = 0, \qquad (31)$$

$$\psi \left(\log \left(1 - \frac{\overline{VaR}}{W} \right)^+ - \left(r + \pi^{*\top} \mu - \frac{1}{2} |\pi^{*\top} \sigma|^2 \right) \tau - N^{-1}(\alpha) |\pi^{*\top} \sigma| \sqrt{\tau} \right) = 0, \quad (32)$$

$$\log\left(1-\frac{\overline{VaR}}{W}\right)^{+} - \left(r+\pi^{*\top}\mu - \frac{1}{2}|\pi^{*\top}\sigma|^{2}\right)\tau - N^{-1}(\alpha)|\pi^{*\top}\sigma|\sqrt{\tau} \le 0.$$
(33)

Rearranging equation (31) gives

$$\left[V_{WW}W^2 - \psi\left(\tau - N^{-1}(\alpha)\frac{\sqrt{\tau}}{|\pi^{*\top}\sigma|}\right)\right]\pi^* = -\left[V_WW + \psi\tau\right](\sigma\sigma^{\top})^{-1}\mu.$$

Since the terms in square brackets are scalar functions of (W, t), this implies that (18) must hold for some scalar function φ . Replacing (18) in (33) gives

$$\log\left(1-\frac{\overline{VaR}}{W}\right)^{+} - \left(r + \left(\varphi - \frac{1}{2}\varphi^{2}\right)|\kappa|^{2}\right)\tau - N^{-1}(\alpha)|\varphi\kappa|\sqrt{\tau} \le 0,$$

which is equivalent to

$$\varphi_{\alpha}^{-} \leq \varphi \leq \varphi_{\alpha}^{+},$$

where φ_{α}^{+} is the function defined in (14) and

$$\varphi_{\alpha}^{-}(W,t) = \frac{|\kappa|\sqrt{\tau} - N^{-1}(\alpha) - \sqrt{(|\kappa|\sqrt{\tau} - N^{-1}(\alpha))^2 - 2\left(\log\left(1 - \frac{\overline{VaR}(W,t)}{W}\right)^+ - r\tau\right)}}{|\kappa|\sqrt{\tau}}$$

Equation (31) and the complementary slackness condition (32) imply

$$(V_{WW}W^2\varphi + V_WW)\mu = 0,$$

or

$$\varphi = -\frac{V_W}{WV_{WW}}$$

when $\varphi_{\alpha}^{-} < \varphi < \varphi_{\alpha}^{+}$, and $\varphi = \varphi_{\alpha}^{\pm}$ otherwise.

Since (18) implies that optimal portfolios are combinations of the riskless asset and the growth-optimal portfolio of risky assets, $(\sigma\sigma^{\top})^{-1}\mu$, the general investment problem (10) can be written equivalently as an investment problem with a single risky asset (the growth-optimal portfolio): because the constraint set for this equivalent problem is convex (see Remark 2), a standard argument implies that the value function V is (increasing and) concave. Hence, the constraint $-V_W/(WV_{WW} > \varphi_{\alpha}^{-}$ is never binding (because φ_{α}^{-} is nonpositive). This establishes the equality in (17).

Finally, (15) follows from substituting (18) and (17) in (30).

PROOF OF LEMMA 1: It is easily verified that $\hat{\rho}_{\hat{\alpha}}(0) = 1 - e^{r\tau} < 0$ and that

$$\frac{\partial}{\partial\varphi}\hat{\rho}_{\hat{\alpha}}(\varphi(\sigma\sigma^{\mathsf{T}})^{-1}\mu) = \exp\left(\left(r+\varphi|\kappa|^{2}\right)\tau\right)\frac{N\left(N^{-1}(\hat{\alpha})-|\varphi||\kappa|\sqrt{\tau}\right)|\kappa|\sqrt{\tau}}{\hat{\alpha}}f(\varphi)$$

for $\varphi \neq 0$, where

$$f(\varphi) = \frac{N'\left(N^{-1}(\hat{\alpha}) - |\varphi| |\kappa| \sqrt{\tau}\right)}{N\left(N^{-1}(\hat{\alpha}) - |\varphi| |\kappa| \sqrt{\tau}\right)} \operatorname{sign}(\varphi) - |\kappa| \sqrt{\tau}.$$

Clearly, $f(\varphi) < 0$ on $(-\infty, 0)$. Moreover, $f(\varphi)$ is monotonically increasing on $(0, +\infty)$, with $\lim_{\varphi \to +\infty} f(\varphi) = +\infty$. Letting

$$\varphi^* = \inf\{\varphi \ge 0 : f(\varphi) \ge 0\},\$$

this implies that $\hat{\rho}_{\hat{\alpha}}(\varphi(\sigma\sigma^{\top})^{-1}\mu)$ is a monotonically decreasing function of φ on $(-\infty, 0)$ and a monotonically increasing function on $(\varphi^*, +\infty)$, with $\hat{\rho}_{\hat{\alpha}}(\varphi(\sigma\sigma^{\top})^{-1}\mu) < 0$ on $[0, \varphi^*]$. Since

mee

$$\lim_{\varphi \to -\infty} \hat{\rho}_{\hat{\alpha}}(\varphi(\sigma\sigma^{\mathsf{T}})^{-1}\mu) = 1 - \lim_{\varphi \to -\infty} \exp\left(\left(r + \varphi|\kappa|^2\right)\tau\right) \frac{N\left(N^{-1}(\hat{\alpha}) + \varphi|\kappa|\sqrt{\tau}\right)}{\hat{\alpha}} = 1$$

and

$$\lim_{\varphi \to +\infty} \hat{\rho}_{\hat{\alpha}}(\varphi(\sigma\sigma^{\top})^{-1}\mu) = 1 - \lim_{\varphi \to +\infty} \exp\left(\left(r + \varphi|\kappa|^2\right)\tau\right) \frac{N\left(N^{-1}(\hat{\alpha}) - \varphi|\kappa|\sqrt{\tau}\right)}{\hat{\alpha}} = 1,$$

the claim immediately follows.

PROOF OF PROPOSITION 2: Letting

$$\varphi^* = \frac{|\kappa|\sqrt{\tau} + N^{-1}(\alpha) + \sqrt{(|\kappa|\sqrt{\tau} + N^{-1}(\alpha))^2 + 2r\tau}}{|\kappa|\sqrt{\tau}}$$

denote the positive root of the equation $\rho_{\alpha}(\varphi(\sigma\sigma^{\top})^{-1}\mu) = 0$, it is easily verified that

$$\frac{\partial}{\partial \varphi} \rho_{\alpha}(\varphi(\sigma\sigma^{\top})^{-1}\mu) = |\kappa| \sqrt{\tau} \Big(1 - \rho_{\alpha}(\varphi(\sigma\sigma^{\top})^{-1}\mu) \Big) \Big(|\kappa| \sqrt{\tau} \varphi - (|\kappa| \sqrt{\tau} + N^{-1}(\alpha)) \Big) > 0$$

for $\varphi > \varphi^*$. Therefore, if α and \overline{VaR} satisfy (26) and (27), we have

$$\overline{VaR}(W,t) = W\rho_{\alpha}(\hat{\varphi}_{\hat{\alpha}}^{+}(W,t)(\sigma\sigma^{\top})^{-1}\mu) \ge W\rho_{\alpha}(\varphi^{*}(\sigma\sigma^{\top})^{-1}\mu) = 0$$

and

$$\varphi_{\alpha}^{+}(W,t) = \sup \left\{ \varphi \ge 0 : W \rho_{\alpha}(\varphi(\sigma\sigma^{\top})^{-1}\mu) \le \overline{VaR}(W,t) \right\} = \hat{\varphi}_{\hat{\alpha}}^{+}(W,t).$$

Similarly, if $\hat{\alpha}$ and \overline{TCE} satisfy (28) and (29), we have

$$\overline{TCE}(W,t) = W\hat{\rho}_{\hat{\alpha}}(\varphi_{\alpha}^{+}(W,t)(\sigma\sigma^{\top})^{-1}\mu) \ge W\hat{\rho}_{\hat{\alpha}}(\Phi_{\hat{\alpha}}^{+}(0)(\sigma\sigma^{\top})^{-1}\mu) = 0$$

and

$$\hat{\varphi}_{\hat{\alpha}}^{+}(W,t) = \sup \left\{ \varphi \ge 0 : W \hat{\rho}_{\hat{\alpha}}(\varphi(\sigma\sigma^{\top})^{-1}\mu) \le \overline{TCE}(W,t) \right\} = \varphi_{\alpha}^{+}(W,t). \qquad \Box$$

PROOF OF COROLLARY 2: When $\overline{VaR}(W,t) = \beta W$, (14) gives $\varphi_{\alpha}^+(W,t) = \varphi^*$ for all $(W,t) \in \mathbb{R}^+ \times [0,T]$, where

$$\varphi^* = \frac{|\kappa|\sqrt{\tau} + N^{-1}(\alpha) + \sqrt{(|\kappa|\sqrt{\tau} + N^{-1}(\alpha))^2 - 2(\log(1-\beta) - r\tau)}}{|\kappa|\sqrt{\tau}}$$

Since (8) implies $\rho_{\alpha}(\varphi(\sigma\sigma^{\top})^{-1}\mu) \leq \hat{\rho}_{\alpha}(\varphi(\sigma\sigma^{\top})^{-1}\mu)$ for all $\varphi \in \mathbb{R}$, we have

$$\inf_{(W,t)\in\mathbb{R}^+\times[0,T]} \varphi_{\alpha}^+(W,t) = \varphi^*$$

$$= \sup\left\{\varphi\in\mathbb{R}: \rho_{\alpha}(\varphi(\sigma\sigma^{\top})^{-1}\mu) \leq \beta\right\}$$

$$\geq \sup\left\{\varphi\in\mathbb{R}: \hat{\rho}_{\alpha}(\varphi(\sigma\sigma^{\top})^{-1}\mu) \leq \beta\right\}$$

$$\geq \sup\left\{\varphi\in\mathbb{R}: \hat{\rho}_{\alpha}(\varphi(\sigma\sigma^{\top})^{-1}\mu) \leq 0\right\}$$

$$= \Phi_{\alpha}^+(0).$$

Therefore, the condition (28) is satisfied with $\hat{\alpha} = \alpha$. It then immediately follows from Proposition 2 that the proportional VaR limit $VaR_t^{\alpha,\pi} \leq \beta W_t^{\pi}$ is equivalent to the proportional TCE limit $TCE_t^{\alpha,\pi} \leq \hat{\beta}W_t^{\pi}$, where $\hat{\beta} = \hat{\rho}_{\alpha}(\varphi^*(\sigma\sigma^{\top})^{-1}\mu)$. Moreover, it follows from (8) that

$$\beta = \rho_{\alpha}(\varphi^*(\sigma\sigma^{\top})^{-1}\mu) \le \hat{\rho}_{\hat{\alpha}}(\varphi^*(\sigma\sigma^{\top})^{-1}\mu) = \hat{\beta}$$

and

$$\hat{\beta} = \hat{\rho}_{\hat{\alpha}}(\varphi^*(\sigma\sigma^{\top})^{-1}\mu) < 1.$$

References

AHN, D.-H., J. BOUDOUKH, M. RICHARDSON AND R.F. WHITELAW, 1999, "Optimal Risk Management Using Options", *Journal of Finance* 54, 359–375.

ALEXANDER, G. AND A. BAPTISTA, 2000, "Economic Implications of Using a Mean-VaR Model for Portfolio Selection: A Comparison with Mean-Variance Analysis", working paper, University of Minnesota.

ARTZNER, P., F. DELBAEN, J.-M. EBER AND D. HEATH, 1999, "Coherent Measures of Risk", *Mathematical Finance* 9, 203–228.

BASAK, S. AND A. SHAPIRO, 2001, "Value-at-Risk Based Risk Management: Optimal Policies and Asset Prices", *Review of Financial Studies* 14, 371–405.

CVITANIĆ, J. AND I. KARATZAS, 1992, "Convex Duality in Constrained Portfolio Optimization", Annals of Applied Probability 2, 767–818.

DUFFIE, D AND J. PAN, 1997, "An Overview of Value at Risk", Journal of Derivatives 4, 7–49.

EMBRECHTS, P., 1999, "Extreme Value Theory as a Risk Management Tool", North American Actuarial Journal.

EMMER, S., C. KLÜPPELBERG AND R. KORN, 2001, "Optimal Portfolios with Bounded Capital at Risk", *Mathematical Finance* **11**, 365–384.

JORION, P., 2001, Value at Risk, McGraw-Hill, New York.

KARATZAS, I. AND S.E. SHREVE, 1988, Brownian Motion and Stochastic Calculus, Springer-Verlag, New York.

KAST, R., E. LUCIANO AND L. PECCATI, 1999, "Value-at-Risk as a Decision Criterion", working paper, University of Turin.

KUSHNER, H.J. AND P.G. DUPUIS, 1992, Numerical Methods for Stochastic Control Problems in Continuous Time, Springer-Verlag, New York.

PFLUG, G., 2000, "Some Remarks on the Value-at-Risk and the Conditional Value-at-Risk", in: S. Uryasev (ed.), Probabilistic Constrained Optimization: Methodology and Applications, Kluwer, Boston.

ROCKAFELLAR, R.T. AND S. URYASEV, 2001, "Optimization of Conditional Value-at-Risk", *Journal of Risk*, forthcoming.

VORST, T., 2001, "Optimal Portfolios under a Value at Risk Constraint", in: C. Casacuberta, R.M. Miró-Roig, J. Verdera and S. Xambó (eds.), *Proceedings of the European Congress of Mathematics, Barcelona, July 10-14, 2000*, Birkhäuser, Basel, forthcoming.

RESEARCH PAPER SERIES

Extra copies of research papers are available to the public upon request. In order to obtain copies of past or future works, please contact our office at the following address: International Center FAME, 40 bd. du Pont d'Arve, Case Postale 3, 1211 Geneva 4. As well, please note that these works are available on our website *www.fame.ch*, under the heading "Research" in PDF format for your consultation. We thank you for your continuing support and interest in FAME, and look forward to serving you in the future.

N° 59: Implicit Forward Rents as Predictors of Future Rents

Peter ENGLUND, Stockholm Institute for Financial Research and Stockholm School of Economics; Åke GUNNELIN, Stockholm Institute for Financial Research; Martin HOESLI University of Geneva (HEC and FAME) and University of Aberdeen (Business School) and Bo SÖDERBERG, Royal Institute of Technology; *October 2002*

N° 58: Do Housing Submarkets Really Matter?

Steven C. BOURASSA, School of Urban and Public Affairs, University of Louisville; Martin HOESLI, University of Geneva (HEC and FAME) and University of Aberdeen (Business School) and Vincent S. PENG, AMP Henderson Global Investors; *November 2002*

N° 57: Nonparametric Estimation of Copulas for Time Series

Jean-David FERMANIAN, CDC Ixis Capital Markets and CREST and Olivier SCAILLET, HEC Genève and FAME, Université de Genève; *November 2002*

N° 56: Interactions Between Market and Credit Risk: Modeling the Joint Dynamics of Default-Free and Defaultable Bond Term Structures Roger WALDER, University of Lausanne, International Center FAME and Banque Cantonale Vaudoise; November 2002

N° 55: Option Pricing With Discrete Rebalancing Jean-Luc PRIGENT, THEMA, Université de Cergy-Pontoise; Olivier RENAULT, Financial Markets Group, London School of Economics and Olivier SCAILLET, HEC Genève and FAME, University of Geneva; July 2002

N° 54: The Determinants of Stock Returns in a Small Open Economy Séverine CAUCHIE, HEC-University of Geneva, Martin HOESLI, University of Geneva (HEC and FAME) and University of Aberdeen (School of Business) and Dušan ISAKOV, HEC-University of Geneva and International

- N° 53: Permanent and Transitory Factors Affecting the Dynamics of the Term Structure of Interest Rates Christophe PÉRIGNON, Anderson School, UCLA and Christophe VILLA, ENSAI, CREST-LSM and CREREG-Axe Finance; *June 2002*
- N° 52: Hedge Fund Diversification: How Much is Enough? François-Serge LHABITANT, Thunderbird University, HEC-University of Lausanne and FAME; Michelle LEARNED, Thunderbird University; *July 2002*
- N° 51: Cannibalization & Incentives in Venture Financing Stefan ARPING, University of Lausanne; *May 2002*

Center FAME; September 2002

N° 50: What Factors Determine International Real Estate Security Returns? Foort HAMELINK, Lombard Odier & Cie, Vrije Universiteit and FAME; Martin HOESLI, University of Geneva (HEC and FAME) and University of Aberdeen; *July 2002*



HEI



- N° 49: Playing Hardball: Relationship Banking in the Age of Credit Derivatives Stefan ARPING, University of Lausanne; *May 2002*
- N° 48: A Geometric Approach to Multiperiod Mean Variance Optimization of Assets and Liabilities Markus LEIPPOLD, Swiss Banking Institute, University of Zurich; Fabio TROJANI, Institute of Finance, University of Southern Switzerland; Paolo VANINI, Institute of Finance, University of Southern Switzerland; April 2002
- N° 47: Why Does Implied Risk Aversion Smile? Alexandre ZIEGLER, University of Lausanne and FAME; *May 2002*
- N° 46: Optimal Investment With Default Risk Yuanfeng HOU, Yale University; Xiangrong JIN, FAME and University of Lausanne; *March 2002*
- N° 45: Market Dynamics Around Public Information Arrivals Angelo RANALDO, UBS Asset Management; *February 2002*
- N° 44: Nonparametric Tests for Positive Quadrant Dependence Michel DENUIT, Université Catholique de Louvain, Olivier SCAILLET, HEC Genève and FAME, University of Geneva; *March 2002*
- N° 43: Valuation of Sovereign Debt with Strategic Defaulting and Rescheduling Michael WESTPHALEN, École des HEC, University of Lausanne and FAME; *February 2002*
- N° 42: Liquidity and Credit Risk Jan ERICSSON, McGill University and Olivier RENAULT, London School of Economics; *August 2001*
- N° 41: Testing for Concordance Ordering Ana C. CEBRIÁN, Universidad de Zaragoza, Michel DENUIT, Université Catholique de Louvain, Olivier SCAILLET, HEC Genève and FAME, University of Geneva; *March 2002*
- N° 40: Immunization of Bond Portfolios: Some New Results Olivier de La GRANDVILLE, University of Geneva; *February 2002*
- N° 39: Weak Convergence of Hedging Strategies of Contingent Claims Jean-Luc PRIGENT, Thema, Université de Cergy-Pontoise; Olivier SCAILLET, HEC Genève and FAME, University of Geneva; *January 2002*
- N° 38: Indirect Estimation of the Parameters of Agent Based Models of Financial Markets Manfred GILLI, University of Geneva; Peter WINKER, International University in Germany; *November 2001*
- N° 37: How To Diversify Internationally? A comparison of conditional and unconditional asset allocation methods. Laurent BARRAS, HEC-University of Geneva, International Center FAME; Dušan ISAKOV, HEC-University of Geneva, International Center FAME; *November 2001*
- N° 36: Coping with Credit Risk Henri LOUBERGÉ, University of Geneva, Harris SCHLESINGER, University of Alabama; October 2001







N° 35: Country, Sector or Style: What matters most when constructing Global Equity Portfolios? An empirical investigation from 1990-2001.

Foort HAMELINK, Lombard Odier & Cie and Vrije Universiteit; Hélène HARASTY, Lombard Odier & Cie; Pierre HILLION, Insead (Singapore), Academic Advisor to Lombard Odier & Cie; *October 2001*

N° 34: Variable Selection for Portfolio Choice

Yacine AÏT-SAHALIA, Princeton University & NBER, and Michael W. BRANDT, Wharton School, University of Pennsylvania & NBER; *February 2001* (Please note: The complete paper is available from the Journal of Finance 56, 1297-1351.)

- N° 33: The Characteristics of Individual Analysts' Forecast in Europe Guido BOLLIGER, University of Neuchâtel and FAME; July 2001
- N° 32: Portfolio Diversification: Alive and Well in Euroland Kpaté ADJAOUTE, HSBC Republic Bank (Suisse), SA and Jean-Pierre DANTHINE, University of Lausanne, CEPR and FAME; *July 2001*
- N° 31: EMU and Portfolio Diversification Opportunities Kpate ADJAOUTÉ, Morgan Stanley Capital International, Geneva and Jean-Pierre DANTHINE, University of Lausanne, CEPR and FAME; *April 2000*
- N° 30: Serial and Parallel Krylov Methods for Implicit Finite Difference Schemes Arising in Multivariate Option Pricing Manfred GILLI, University of Geneva, Evis KËLLEZI, University of Geneva and FAME, Giorgio PAULETTO, University of Geneva; March 2001
- N° 29: Liquidation Risk

Alexandre ZIEGLER, HEC-University of Lausanne, Darrell DUFFIE, The Graduate School of Business, Stanford University; *April 2001*

- N° 28: Defaultable Security Valuation and Model Risk Aydin AKGUN, University of Lausanne and FAME; *March 2001*
- N° 27: On Swiss Timing and Selectivity: in the Quest of Alpha François-Serge LHABITANT, HEC-University of Lausanne and Thunderbird, The American Graduate School of International Management; *March 2001*
- N° 26: Hedging Housing Risk

Peter ENGLUND, Stockholm School of Economics, Min HWANG and John M. QUIGLEY, University of California, Berkeley; *December 2000*

- N° 25: An Incentive Problem in the Dynamic Theory of Banking Ernst-Ludwig VON THADDEN, DEEP, University of Lausanne and CEPR; December 2000
- N° 24: Assessing Market Risk for Hedge Funds and Hedge Funds Portfolios François-Serge LHABITANT, Union Bancaire Privée and Thunderbird, the American Graduate School of International Management; *March 2001*



HEI



- N° 23: On the Informational Content of Changing Risk for Dynamic Asset Allocation Giovanni BARONE-ADESI, Patrick GAGLIARDINI and Fabio TROJANI, Università della Svizzera Italiana; March 2000
- N° 22: The Long-run Performance of Seasoned Equity Offerings with Rights: Evidence From the Swiss Market Michel DUBOIS and Pierre JEANNERET, University of Neuchatel; *January 2000*
- N° 21: Optimal International Diversification: Theory and Practice from a Swiss Investor's Perspective Foort HAMELINK, Tilburg University and Lombard Odier & Cie; *December 2000*
- N° 20: A Heuristic Approach to Portfolio Optimization Evis KËLLEZI; University of Geneva and FAME, Manfred GILLI, University of Geneva; October 2000
- N° 19: Banking, Commerce, and Antitrust Stefan Arping; University of Lausanne; *August 2000*
- N° 18: Extreme Value Theory for Tail-Related Risk Measures Evis KËLLEZI; University of Geneva and FAME, Manfred GILLI, University of Geneva; October 2000
- N° 17: International CAPM with Regime Switching GARCH Parameters Lorenzo CAPIELLO, The Graduate Institute of International Studies; Tom A. FEARNLEY, The Graduate Institute of International Studies and FAME; *July 2000*
- N° 16: Prospect Theory and Asset Prices Nicholas BARBERIS, University of Chicago; Ming HUANG, Stanford University; Tano SANTOS, University of Chicago; *September 2000*
- N° 15: Evolution of Market Uncertainty around Earnings Announcements Dušan ISAKOV, University of Geneva and FAME; Christophe PÉRIGNON, HEC-University of Geneva and FAME; *June 2000*
- N° 14: Credit Spread Specification and the Pricing of Spread Options Nicolas MOUGEOT, IBFM-University of Lausanne and FAME; *May 2000*
- N° 13: European Financial Markets After EMU: A First Assessment Jean-Pierre DANTHINE, Ernst-Ludwig VON THADDEN, DEEP, Université de Lausanne and CEPR and Francesco GIAVAZZI, Università Bocconi, Milan, and CEPR; *March 2000*
- N° 12: Do fixed income securities also show asymmetric effects in conditional second moments? Lorenzo CAPIELLO, The Graduate Institute of International Studies; *January 2000*
- N° 11: Dynamic Consumption and Portfolio Choice with Stochastic Volatility in Incomplete Markets George CHACKO, Harvard University; Luis VICEIRA, Harvard University; September 1999
- N° 10: Assessing Asset Pricing Anomalies Michael J. BRENNAN, University of California, Los Angeles; Yihong XIA, University of Pennsylvania; *July 1999*



HEI



N° 9: Recovery Risk in Stock Returns

Aydin AKGUN, University of Lausanne & FAME; Rajna GIBSON, University of Lausanne; July 1999

- **N° 8: Option pricing and replication with transaction costs and dividends** Stylianos PERRAKIS, University of Ottawa; Jean LEFOLL, University of Geneva; *July1999*
- N° 7: Optimal catastrophe insurance with multiple catastrophes Henri LOUBERGÉ, University of Geneva; Harris SCHLESINGER, University of Alabama; *September 1999*
- N° 6: Systematic patterns before and after large price changes: Evidence from high frequency data from the Paris Bourse Foort HAMELINK, Tilburg University; *May 1999*
- N° 5: Who Should Buy Long-Term Bonds? John CAMPBELL, Harvard University; Luis VICEIRA, Harvard Business School; *October 1998*
- N° 4: Capital Asset Pricing Model and Changes in Volatility André Oliveira SANTOS, Graduate Institute of International Studies, Geneva; *September 1998*
- N° 3: Real Options as a Tool in the Decision to Relocate: An Application to the Banking Industry Pascal BOTTERON, HEC-University of Lausanne; *January 2000*
- N° 2: Application of simple technical trading rules to Swiss stock prices: Is it profitable? Dušan ISAKOV, HEC-Université de Genève; Marc HOLLISTEIN, Banque Cantonale de Genève; *January 1999*
- **N° 1:** Enhancing portfolio performance using option strategies: why beating the market is easy. François-Serge LHABITANT, HEC-University of Lausanne; *December 1998*





International Center FAME - Partner Institutions

The University of Geneva

The University of Geneva, originally known as the Academy of Geneva, was founded in 1559 by Jean Calvin and Theodore de Beze. In 1873, The Academy of Geneva became the University of Geneva with the creation of a medical school. The Faculty of Economic and Social Sciences was created in 1915. The university is now composed of seven faculties of science; medicine; arts; law; economic and social sciences; psychology; education, and theology. It also includes a school of translation and interpretation; an institute of architecture; seven interdisciplinary centers and six associated institutes.

More than 13'000 students, the majority being foreigners, are enrolled in the various programs from the licence to high-level doctorates. A staff of more than 2'500 persons (professors, lecturers and assistants) is dedicated to the transmission and advancement of scientific knowledge through teaching as well as fundamental and applied research. The University of Geneva has been able to preserve the ancient European tradition of an academic community located in the heart of the city. This favors not only interaction between students, but also their integration in the population and in their participation of the particularly rich artistic and cultural life. *http://www.unige.ch*

The University of Lausanne

Founded as an academy in 1537, the University of Lausanne (UNIL) is a modern institution of higher education and advanced research. Together with the neighboring Federal Polytechnic Institute of Lausanne, it comprises vast facilities and extends its influence beyond the city and the canton into regional, national, and international spheres.

Lausanne is a comprehensive university composed of seven Schools and Faculties: religious studies; law; arts; social and political sciences; business; science and medicine. With its 9'000 students, it is a medium-sized institution able to foster contact between students and professors as well as to encourage interdisciplinary work. The five humanities faculties and the science faculty are situated on the shores of Lake Leman in the Dorigny plains, a magnificent area of forest and fields that may have inspired the landscape depicted in Brueghel the Elder's masterpiece, the Harvesters. The institutes and various centers of the School of Medicine are grouped around the hospitals in the center of Lausanne. The Institute of Biochemistry is located in Epalinges, in the northern hills overlooking the city. *http://www.unil.ch*

The Graduate Institute of International Studies

The Graduate Institute of International Studies is a teaching and research institution devoted to the study of international relations at the graduate level. It was founded in 1927 by Professor William Rappard to contribute through scholarships to the experience of international co-operation which the establishment of the League of Nations in Geneva represented at that time. The Institute is a self-governing foundation closely connected with, but independent of, the University of Geneva.

The Institute attempts to be both international and pluridisciplinary. The subjects in its curriculum, the composition of its teaching staff and the diversity of origin of its student body, confer upon it its international character. Professors teaching at the Institute come from all regions of the world, and the approximately 650 students arrive from some 60 different countries. Its international character is further emphasized by the use of both English and French as working languages. Its pluralistic approach - which draws upon the methods of economics, history, law, and political science -reflects its aim to provide a broad approach and in-depth understanding of international relations in general. *http://heiwww.unige.ch*



HEL





40, Bd. du Pont d'Arve PO Box, 1211 Geneva 4 Switzerland Tel (++4122) 312 09 61 Fax (++4122) 312 10 26 http: //www.fame.ch E-mail: admin@fame.ch



